

Cadmium Pollution From Zinc-Smelters up to Fourfold Higher Than Expected in Western Europe in the 1980s as Revealed by Alpine Ice

Michel Legrand, Joe Mcconnell, L. Lestel, Susanne Preunkert, Monica Arienzo, Nathan Chellman, Andreas Stohl, Sabine Eckhardt

► To cite this version:

Michel Legrand, Joe Mcconnell, L. Lestel, Susanne Preunkert, Monica Arienzo, et al.. Cadmium Pollution From Zinc-Smelters up to Fourfold Higher Than Expected in Western Europe in the 1980s as Revealed by Alpine Ice. Geophysical Research Letters, 2020, 47 (10), 10.1029/2020GL087537. hal-03039001

HAL Id: hal-03039001 https://hal.science/hal-03039001

Submitted on 11 Dec 2020 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

1	
2	
3	Cadmium pollution from zinc-smelters up to four-fold higher than expected in
4	western Europe in the 1980s as revealed by alpine ice
5 6	M. Legrand ¹ , J. R. McConnell ² , L. Lestel ³ , S. Preunkert ¹ , M. Arienzo ² , N. J. Chellman ² , A. Stohl ^{4,5} , and S. Eckhardt ⁴
7 8	¹ CNRS, Institut des Géosciences de l'Environnement, Université Grenoble Alpes, Grenoble, France.
9	² Division of Hydrologic Sciences, Desert Research Institute, Reno, Nevada, USA.
10	³ Sorbonne Université-CNRS-EPHE, UMR 7619 METIS, Paris, France.
11 12	⁴ Department of Atmospheric and Climate Research, Norwegian Institute for Air Research, Kjeller, Norway.
13	⁵ Now at Department of Meteorology and Geophysics, University of Vienna, Vienna, Austria.
14	
15	Corresponding author: Michel Legrand (michel.legrand@univ-grenoble-alpes.fr)
16	
17	Key Points:
18 19	• First evaluation of past cadmium and lead anthropogenic emissions in Europe by using alpine ice records and model simulations over the Alps
20 21	• Maximum of cadmium pollution in Europe in 1980 instead of 1965 due to less efficient reduction of emissions than predicted
22 23 24	• Maximum pollution by cadmium from zinc smelters and by lead from leaded gasoline in the late 1970s in Europe

25 Abstract

- 26 Estimates of past emission inventories suggest that toxic heavy metal pollution in Europe was
- highest in the mid-1970s for lead and in the mid-1960s for cadmium, but these previous
- estimates have not been compared to observations. Here, alpine ice-cores were used to document
- cadmium and lead pollution in western Europe between 1890 and 2000. The ice-core trends
- 30 show that while lead pollution largely from leaded gasoline reached a maximum in \sim 1975 as
- expected, cadmium pollution primarily from zinc smelters peaked in the early-1980s rather than
- in ~1965, and was up to four-fold higher than estimated after 1975. Comparisons between ice-
- 33 core trends, estimated past emissions, and state-of-the-art atmospheric aerosol transport and
- 34 deposition modelling suggest that the estimated decreases in cadmium emissions after 1970 were 35 based on overly optimistic emissions reductions from the introduction of pollution control
- 36 devices and other technological improvements.

37 Plain Language Summary

Cadmium and lead are among the most toxic heavy-metal pollutants, and identified by
 international conventions as priority contaminants for emissions reduction. Anthropogenic

- emissions in Europe previously estimated from commodity production statistics are uncertain;
- 40 emissions in Europe previously estimated from commonly production statistics are uncertain
 41 comparison with long-term pollution records extracted from well-dated ice cores provides a
- 41 comparison with long-term ponution records extracted from wen-dated ice cores provides a 42 means of evaluating these estimates. Alpine ice-core records spanning 1890 to 2000 show that
- lead pollution, mainly related to leaded gasoline use since the mid-20th century, reached a
- maximum in the mid-1970s as predicted by estimated past emissions. Surprisingly, however,
- 44 maximum in the initial provide as predicted by estimated past emissions. Surprisingly, however,
 45 cadmium levels primarily arising from zinc smelter emissions were up to four-fold higher than
- 46 expected and decreased only after 1980, suggesting that previous reconstructions of past
- cadmium pollution had assumed overly optimistic emissions reductions from the introduction of
- 48 pollution control devices and other technological improvements particularly after 1975. Contrary
- 49 to previous emissions estimates, these new alpine ice-core records show that Western European
- 50 pollution maxima for these two toxic metals coincided in the 1970s. This finding is important in
- 51 terms of impact of Cd pollution on organisms that depends on concentration in the environment
- 52 but also duration of exposure.

53 **1 Introduction**

The atmospheric concentrations and deposition of heavy metals have increased since pre-54 industrial times because of human activities, sometimes causing adverse health effects in humans 55 as well as wildlife and the environment (Nriagu, 1990). Among toxic metals, cadmium (Cd), lead 56 (Pb), and mercury (Hg) were identified by international conventions as priority contaminants 57 (Vestreng et al., 2006). Based on statistical information and emission factors (EFs) estimated for 58 relevant emission source categories (fossil fuel and ore consumption, production of various types 59 of industrial goods), global anthropogenic emissions were calculated for 1983 and the mid-1990s 60 (Ngiaru & Pacyna, 1988; Pacyna & Pacyna, 2001). At the scale of Europe, several studies 61 estimated anthropogenic emissions every 5 to 10 years from 1955 to 2000 (Olendrzynski et al., 62 1996; Pacyna & Pacyna, 1999; Pacyna et al., 2009), but little is known prior to 1955. Air and 63 precipitation monitoring, available in Europe since 1990 for Pb and Cd (Pacyna et al., 2009), 64 permit evaluation of post-1990 declines in emissions. However, no direct atmospheric 65 observations are available either for the 1965-1975 period thought to correspond to peak 66 emissions (Pacyna et al., 2009) or to evaluate expected decreases following implementation of 67 68 various policies in Europe to reduce emissions. Moreover, no inventories exist for Europe from

69 1955 back to the onset of the industrial period (~1850). Evaluation of anthropogenic

- 70 perturbations also requires understanding of natural sources, and estimates developed through
- extrapolation of sparse datasets have large uncertainties (e.g., a factor of 20 for Pb and Cd
- 72 (Nriagu, 1979)). In its 2015 status report, the European Monitoring and Evaluation Programme
- 73 (EMEP) concluded that uncertainties in available emission data are still significant for recent
- 74 decades and there is a need to better quantify the contributions of anthropogenic versus natural 75 sources at remote sites to develop further strategies for emission reductions (Ilyin et al., 2015).

Except for Hg that is poorly scavenged from the atmosphere by precipitation, well-dated 76 ice-cores have been studied to evaluate past deposition of numerous metals and thus can help to 77 reduce emissions uncertainties. Records from Greenland ice-cores documented past pollution 78 from growing anthropogenic emissions of Pb and Cd from industrialized regions of the Northern 79 Hemisphere, particularly North America (McConnell & Edwards, 2008). More intimately 80 connected to European emissions, alpine ice records have been used to evaluate several aspects 81 of pollution over Europe, with most studies conducted at Colle Gnifetti (Monte Rosa, CG) 82 (Schwikowski, 2004; Wagenbach et al., 1988; 2012) and Col du Dome (Mont Blanc, CDD) (83

84 Preunkert & Legrand, 2013).

The longest Alpine ice-core records have been obtained at CG due to low snow 85 accumulation rate, with continuous profiles of Pb and Cd extending back to the Middle Ages 86 (Gabrieli & Barbante, 2014; Gabrieli, 2008; Schwikowski et al., 2004). These records identified 87 a large increase of pollution between \sim 1960 and 1980 and a subsequent recent (1980 to 1995) 88 89 decrease. Given the larger pollution deposition in summer relative to winter (a factor of 5 to 10, Preunkert et al., 2002) when vertical atmospheric mixing is strong, accurate interpretation of ice 90 91 records extracted from a low snow accumulation site is limited by large uncertainties in the seasonal variations in snow deposition and/or preservation affected by preferential wind scouring 92 of winter snowfall. At CDD, winter snow scouring occurs but the high snow accumulation rate at 93 the site enables distinguishing between summer and winter layers in the ice dating back to the 94 95 past century (Wagenbach et al., 2012). Prior to this study, no seasonally-resolved ice records of metals have been reported from the CDD site. 96

We developed a seasonally-resolved continuous ice-core record of Pb, Cd, and Zn concentrations spanning the 20th century from ice-cores extracted at CDD. We targeted the toxic heavy metals Pb and Cd, as well as Zn as an indicator of emissions from zinc-smelters, which also are prominent sources of Cd emissions. We interpreted these ice-core records in the context of reported European emissions and statistics of various source categories together with state-ofthe-art FLEXPART atmospheric transport and deposition modelling.

103 2 Materials and Methods

104 2.1 Ice-core material and Dating

105 Chemical measurements were made on two ice-cores extracted at Col du Dome (4,250 m 106 above sea level, French Alps). Melting of surface snow occasionally occur at the site, however, 107 the thickness of the infiltration ice layers does not exceed 0.2 m, which is small compared to the 108 annual layer thickness of 1.25 to 4.35 meters water equivalent (Preunkert et al., 2000). The C10 109 core (126 m long) was drilled in 1994 and the CDM core (140 m long), extracted less than 20 m 110 away from the C10, was drilled in 2012. Initially allocated to the study of metals, halogens, and 111 organic carbon, the analysis were restricted to the ice portion of the core to avoid potential contamination of porous firn portion of the core by organics (Legrand et al., 2013) and loss of
iodine from the firn after core extraction (Legrand et al., 2018). In addition to analyzing the ice
portion of the C10 core (55.0 m to 125.7 m depth), we extended the time period to span the years
between 1983-2000 (i.e., 46.1 to 78.0 m depth) in the CDM core.

The upper sections of both cores were dated by annual layer counting primarily using 116 pronounced seasonal variations in ammonium concentrations (summer to winter ratio exceeded 117 10, Fig. S1). Winter layers generally thin with depth relative to summer layers because of 118 changes in depositional processes upstream of the CDD drilling site, although we were able to 119 identify winter layers for annual layer counting back to 1890. However, many of these winter 120 ammonium minima were too thin to reliably determine winter concentrations of Pb, Cd, and Zn 121 so prior to 1930, only five winter values were calculated (Fig. 1). Note also that during the 122 1960s, samples corresponding to full winter (based on ammonium values) were rather sparse and 123 eight winter values were not calculated. The resulting chronology was assigned an age of 1890 at 124 118.3 m (Legrand et al., 2018). Prior to 1890, no winter snow was preserved, but the age of the 7 125 lowest meters (below 118.3 m depth) of the C10 core were estimated to cover at least the period 126 1830-1890 (i.e., prior to the onset of the pre-industrial period in ~1850) (Text S1). Past summer 127 trends including during the pre-industrial time period are thus better documented so in the 128 following we evaluate and discuss mainly summer trends. 129

130

2.2 Measurements of metals and comparison with previous studies

CDD ice-cores were analyzed for a broad range of elements using the continuous flow 131 132 ice-core analytical system at the Desert Research Institute (DRI) Ultra Trace Chemistry Laboratory (McConnell et al., 2019). Longitudinal samples $(3.3 \times 3.3 \text{ cm cross section})$ were 133 melted sequentially on a heated ceramic melter head and the meltwater stream split into three 134 regions by ridges engraved in the melter head. Pb, Cd, Zn, and co-analyzed species such as 135 cerium (Ce) were measured in the meltwater from the innermost ring (10% of the sample 136 volume) using two high-resolution inductively coupled plasma-mass spectrometers (HR-ICP-137 MS) operating in parallel. The detection limits, defined as 3 times the standard deviation of the 138 blank, were 0.0004 ng g^{-1} for Pb, 0.0001 ng g $^{-1}$ for Cd, 0.009 ng g $^{-1}$ for Zn, and 0.0003 ng g $^{-1}$ for 139 Ce. Our new CDD measurements are in excellent agreement with previously-published, discrete 140 measurements performed on CDD ice-cores (Table S1). Comparison with those previously 141 obtained at CG indicates slightly higher values at CG than at CDD, likely due to less winter 142 snow preservation at CG than at CDD (Wagenbach et al., 2012). 143

144 2.3 The crustal fraction of Zn, Cd, and Pb

Many metals are naturally emitted within the atmosphere along with soil and desert dust 145 particles. The crustal contribution to each metal can be estimated from Ce measurements using 146 the "mean sediment" metal-to-Ce ratio (Bowen, 1966). Measurement of total concentrations of 147 Ce that is part of relatively insoluble dust is, however, influenced by acidification times because 148 149 Ce remains in the particle phase longer than other elements associated with pollution that are adsorbed onto small particles and are readily washed from the outside of the particles during on-150 line acidification of the continuous analytical system. Previous assessment of measurement 151 recovery during continuous measurements with the DRI system indicated that recovery was 100 152 % and 60% for Pb and Ce, respectively (McConnell et al., 2018). We assume a similarly high 153 recovery for Zn and Cd since they are also mainly associated with pollution. 154

Since the metal/Ce ratio can deviate from the "mean sediment" value and so result in an 155 incorrect determination of the crustal component, we used an alternative approach to estimate 156 site-specific metal/Ce ratios using ice samples containing relatively high dust or low 157 anthropogenic contributions. First, in ice samples deposited over the recent decades we used 158 alkalinity to select samples containing high Saharan dust concentrations (Wagenbach et al., 159 1996). Second, we assumed that non-crustal contributions were negligible in some samples 160 deposited during pre-industrial time. Using both approaches, relationships between the metals 161 and Ce in dust-dominated CDD samples suggest site-specific crustal ratios of 1.9 for Zn/Ce, 0.02 162 for Cd/Ce, and ~0.4 for Pb/Ce (Fig. S3, Text S2). After correcting the Ce concentrations for the 163 60% under-recovery, these site-specific crustal ratios are 1.12 for Zn compared to 1.15 in mean 164 sediment, 0.012 instead of 0.002 for Cd, and 0.23 (i.e., similar to the mean sediment value) for 165 Pb. 166

167 2.4 FLEXPART and metal deposition fluxes at CDD

To compare the CDD ice-core summer trends with estimated past emissions in relevant 168 countries we used the Lagrangian particle dispersion model FLEXPART by applying a recent 169 development that allows running the model backward in time for dry and wet deposition (Stohl 170 et al., 2005; Eckhardt et al., 2017). The model maps the sensitivity of deposition at the CDD site 171 to emission flux. The model was run in backward mode at monthly intervals for the period 1901 172 through 1999, and particles were traced backward for 30 days. As meteorological input to 173 FLEXPART, we used the recently completed coupled climate reanalysis for the 20th century 174 (CERA-20C) (Laloyaux et al., 2018) performed at the European Centre for Medium Range 175 Weather Forecasts at a resolution of $2^{\circ} \times 2^{\circ}$ and every six hours. As metal aerosol produced by 176 pollution consists of fine particles (diameter range of 0.5-1.2 µm, Nriagu, 1990) the model was 177 run for 0.4 µm diameter aerosol. Average summer emission sensitivity reported in Fig. S4 were 178 used to calculate summer deposition at CDD. Changing the aerosol size from 0.4 µm to 1 µm 179 leads to a decrease by 30% of emission sensitivities in the countries surrounding the Alps (Fig. 180 S5) and the calculated depositions at CDD are similarly reduced but the long-term temporal 181 trends remain essentially identical. 182

183 2.5 Past emissions in Europe

Past anthropogenic emissions of Pb, Cd, and Zn already were estimated for individual 184 European countries (Olendrzynski et al., 1996; Pacyna & Pacyna, 1999; Pacyna et al., 2009) for 185 the 1955-2000 time period, based on statistical data, EF values, and their changes over time. 186 While accurate information is available from European national authorities to quantify the 187 reduction of Pb additive in gasoline, there is much more uncertainty associated with the decrease 188 of EFs for non-ferrous smelters as well as the iron and steel industry during the 1980s (Pacyna 189 1991). The smelting-related EFs are assumed to have decreased with the progressive installation 190 of electrostatic precipitators and filters in the mid-1970s, initial flue gas desulfurization (FGD) 191 192 installations in the mid-1980s, and further deployment of FGD in the 1990s. EFs proposed for the 1950s and 1960s (Table S2) are estimated to have decreased in the 1980s by a factor of 3 for 193 non-ferrous smelters and a factor of 5 for iron and steel manufacturing (Pacyna, 1991). For Cd 194 emissions from Zn smelters, an EF value of 1500 g per ton of zinc produced has been proposed 195 as an upper limit. When using the imperial smelter furnace (ISF) process, EF drops to 50 g per 196 ton of zinc produced. We have limited information on the use of this technique in various 197 198 European countries over the recent past. Since 1986, ~33% of production in France and 20% in

Germany used ISF, whereas ISF was not used either in Spain or in Belgium (Minemet, 1986). In
 our calculations we therefore have applied an average EF value of 1200 g of Cd per tons of zinc
 produced.

To extend the reconstruction prior to 1955 and discuss some discrepancies between ice-202 core trends and past deposition at CDD calculated by FLEXPART from estimated 1955-2000 203 204 emissions, we calculated past emissions using statistics of relevant sources categories in European countries. We used annual coal consumption, as well as pig iron and steel production 205 statistics available prior to 1975 (Mitchell, 1975). For the 1975-2000 period, we used coal 206 consumption from British Petroleum (http://tools.bp.com/energy-charting-tool). For non-ferrous 207 metallurgy (primary and secondary productions from Zn, Pb and Cu smelters, Pb mining), as 208 well as pig iron and steel production, we used statistical data from (Schmitz, 1979). For the more 209 recent years we used data from Minéraux et métaux statistiques and U.S. Geological Survey 210 statistics. We have assumed EF values reported for prior to 1970 as detailed in Table S2. For Pb, 211 statistical data on leaded gasoline emissions derived from (Pacyna & Pacyna, 1999; Pacyna et 212 al., 2009) were used here since the past amount of lead additive is fairly well documented for 213 each country. For leaded gasoline emissions prior to 1955, we scaled down emissions from the 214 different countries reported in Pacyna et al. (2009) using statistical data on oil consumption from 215 (Mitchell, 1975). Since we focused on summer changes, we have considered the seasonality of 216 217 emissions from coal burning that are about double in winter compared to summer in most European countries (except in the south), whereas emissions from all other sources can be 218 assumed to be constant throughout the year (Fagerli et al., 2007). 219

220 **3 Results and Discussions**

221

3.1 High-resolution alpine ice record of pollution by lead, cadmium, and zinc

Sub-annually resolved measurements of Pb, Cd, and Zn in alpine ice provide complete emission records over the 20th century. To isolate variations in the ice record caused by anthropogenic emissions, we subtracted crustal contributions determined using cerium (McConnell and Edwards, 2008). Such corrections account for an increase of the crustal material deposited at CDD, particularly after 1960 (Preunkert & Legrand, 2013). On average between 1890 and 2000, the crustal contributions comprised of the total 1% for Pb, 6% for Cd, and 9% for Zn.

The quasi-continuous (1890-2000) records of non-crustal Pb (ncPb), Cd (ncCd), and Zn 229 (ncZn) are reported in Fig. 1 together with mean summer and winter levels. The use of annual 230 mean (M-yr) ice concentrations, as commonly done in previous studies of metals in alpine ice-231 232 cores, may bias the atmospheric signal because of year-to-year changes in winter relative to summer snow deposition and preservation. This is particularly true for metals that exhibit 233 234 pronounced seasonal contrast (typical summer to winter ratios are 5 to 10, Fig. S1). To avoid the potential bias of using averaged M-yr concentrations, ice-core trends can be compared to past 235 236 emissions on a seasonal basis (summer and winter) or using annual arithmetic means calculated using the respective summer and winter level in each year (M-W/S). As seen in section 2.1, past 237 summer trends including during the pre-industrial are better documented and in the following we 238 evaluate and discuss mainly summer trends. 239

The summer pre-industrial values, are close to 0.001 ng g^{-1} for ncCd and 0.080 ng g^{-1} for ncZn (prior to ~1875), and 0.025 ng g^{-1} for ncPb (pre- and post-Roman period) (Table S3). The

three times higher pre-industrial level of ncZn compared to ncPb is very consistent with

estimates of global natural emissions of non-crustal Zn (26 ktons a^{-1}) and Pb (8 ktons a^{-1})

244 (Ngiaru, 1979). Global natural emissions of non-crustal Cd are far lower (1.1 ktons a⁻¹) (Ngiaru,

1979). The difference by a factor of ~25 in global natural emissions of ncZn and ncCd is lower,
however, than the difference in ice pre-industrial levels (a factor of 80). Since natural non-crustal

emissions of Cd are dominated by volcanic emissions (0.8 ktons a^{-1} out of a total of 1.1 ktons a^{-1}

¹), it is possible that while Cd volcanic emissions are dominant at a global scale, they have a

249 weaker impact on Europe.

The ncPb summer levels that already largely exceeded its pre-industrial ice level (< 0.1250 ng g^{-1} , Table S3) at the beginning of the 20th century slightly rose from 0.8 ng g^{-1} in 1890 to 1.2 251 ng g⁻¹ in 1950 (Fig. 1). After 1950, summertime ncPb levels rapidly increased to a maximum of 252 ~2.6 ng g⁻¹ at the beginning of the 1970s, after which levels sharply decreased over the following 253 three decades, the level in 2000 ($\sim 0.5 \text{ ng g}^{-1}$) being almost half of those observed at the 254 beginning of the 20th century. The long-term trends of ncCd and ncZn are similar, but differ from 255 ncPb, with a smaller relative increase above the pre-industrial background (0.001 ng g^{-1} to 0.005 256 ng g^{-1} for ncCd) at the beginning of the century, and a less marked decrease after the 1970s, with 257 levels remaining higher (~0.008 ng g^{-1} for ncCd) in 2000 than at the beginning of the century. 258 Note that decreased preservation of winter snow in the 1960s (section 2.1) resulted in maximum 259 M-yr values in ~1965 for all three species, in contrast with the maximum M-W/S means (similar 260 to summer means) for ncPb and ncCd that indicate maxima in ~1975 and ~1980, respectively 261 (Fig. S5). 262

263

3.2 Past emissions of Pb, Cd, and Zn in European countries and deposition at CDD

Past metal deposition fluxes at CDD were calculated from estimated past emissions in European countries that were weighted by the emission sensitivities (averaged for each country) of the site calculated by the Lagrangian particle dispersion model FLEXPART. Average summer emission sensitivities calculated by FLEXPART (Fig. S4) indicate that CDD is most sensitive to emissions originating from countries surrounding the Alps, in decreasing order Switzerland (CH), France (FR), Italy (IT), Spain (ES), ex West Germany (W-DE), Belgium (BE), and United Kingdom (UK).

Three emissions scenarios were used to estimate past depositional fluxes at CDD by scaling emission estimates for each country by the country's average FLEXPART emissions sensitivity. The first scenario estimates fluxes for recent decades using previously published country-specific metal emissions from 1955 to the 1980s (Olendrzynski et al., 1996; Pacyna & Pacyna, 1999) and for 2000 (Pacyna et al., 2009) on the basis of statistical data and considering successive reductions of emission factors (EFs) beginning in the 1970s (Fig. 2; red squares denoted Scenario [1]).

The second scenario estimates depositional fluxes back to 1850 (Fig. 2; blue triangles 278 denoted Scenario [2]) using statistical data of relevant source categories for Pb, Cd, and Zn 279 emissions including smelters (lead, zinc, and copper), pig iron and steel manufacturing, and coal 280 281 consumption available since 1850. For these source categories, we have used the same EFs as for 1955-1970 (Table S2). For Pb emissions, we also considered lead mining and gasoline 282 consumption. Since Pb emissions from leaded gasoline are accurately known based on applied 283 regulations in each European country, we used the Pb emissions from leaded gasoline calculated 284 285 for each country for 1955 (i.e., the same as in Scenario [1]).

The third scenario also calculates metal deposition for 1970 to 2000 using statistical 286 emissions data from Scenario [1], but uses pre-emissions-control EFs (i.e. does not account for 287 reduction in EFs associated with emissions regulation since 1970) (Fig. 2; Emissions Scenario 288 [3]). As in scenarios [1] and [2], we used the Pb emissions from gasoline calculated for each 289 country in (Pacyna & Pacyna, 1999; Pacyna et al., 2009). As seen in Fig. 2, for recent decades 290 the three emissions scenarios are consistent for lead, as the three scenarios all use the same 291 leaded gasoline emissions estimates. For Zn and Cd, a good agreement between scenario [1] and 292 [2] is observed from 1955 to 1970 as expected. The large difference between scenario [1] and [3] 293 after 1970 is due to the difference in EFs, and shows how reduced EFs from non-ferrous smelters 294 (Pb, Zn, and Cu) as well as iron and steel industry after 1970 counteracted increased productions 295 296 from most source categories (except coal consumption that decreased by a factor of 2 from 1965 297 to 2000).

The estimated deposition of anthropogenic Pb, Cd, and Zn at CDD partitioned by source 298 at the beginning of the 20th century and in 1955 to 1970 are detailed in Fig. S6. For 299 anthropogenic Pb, the main emissions were Zn and Pb smelting at the beginning of the 20th 300 century, and leaded gasoline in the 1960s and 1970s. At the start of the 20th century, Pb 301 deposition from Zn smelting was approximately evenly split between FR, BE, and W-DE, as the 302 \sim 3-4 times lower Zn production in FR compared to BE and W-DE was compensated for by its 303 304 much higher emissions sensitivity (Table S4). Contributions from Pb smelting were dominated by ES, with lesser contributions from W-DE and FR. In 1955 to 1970, Pb deposition from leaded 305 gasoline emissions was primarily sourced from FR, followed by IT. For anthropogenic Zn and 306 Cd, Zn smelting dominated emissions since 1890 (Fig. S6), though country-specific 307 contributions evolved over the 20th century. Zn smelting contributions to both Zn and Cd 308 deposition were evenly split between FR, BE, and W-DE at the beginning of the 20th century, but 309 dominated by FR from 1955 to 1970 (Table S4). BE followed by FR became the largest Zn 310 producers over recent decades. 311

It is not possible to directly compare calculated FLEXPART deposition fluxes to ice-core concentrations because of uncertainties associated with precipitation rates simulated by the coarse-resolution model (220 km x 155 km for the CDD grid cell), partly related to the fact that the Alpine orography is not well resolved, and because the annual layer ice thicknesses, which decrease with depth due to ice thinning, do not directly represent past precipitation rates. Nevertheless, relative changes of ice concentrations should be related to past changes in emissions and EFs.

319 3.3 The 1890-2000 lead and cadmium pollution

It is first interesting to note that the calculated deposition of ncPb and ncZn (~5 mg m⁻² a⁻¹ in 1965) are ~50-70 times higher than ncCd deposition (~0.1 mg m⁻² a⁻¹ in 1965), consistent with the ~70 times higher measured ice concentrations of ncPb and ncZn (~2 ng g⁻¹ in 1965) relative to ncCd (~0.3 ng g⁻¹) (Fig. 2).

We begin the comparison between ice-core trends and past emissions for the most recent time-period over which more information on emissions are available. The ncPb summer ice trend shows a rapid increase of lead pollution over the second half of the 20th century that reached a maximum in ~1975 before subsequently decreasing by a factor of 6 by 2000 (Fig. 2a). The changes observed in the CDD ice-core are consistent with estimates of past anthropogenic Pb deposition at CDD that were dominated by leaded gasoline emissions (almost 70% in 1965), with smaller contributions from non-ferrous smelters (~15% in 1965), pig iron and steel (10%,

- mainly from pig iron), and coal (4%) (Fig. S6). The post-1975 ncPb decrease is attributed to the
- reduction of Pb additives to gasoline that reduced gasoline-derived Pb emissions despite the
- continuous growth of gasoline consumption. Using the same leaded gasoline emissions, the two
- estimates of Pb deposition (triangles and squares in Fig. 2) agree well during the rapid increases of emissions related to leaded gasoline. As leaded gasoline emissions declined during the mid-
- 1990s, calculations assuming pre-emissions control EFs for non-ferrous smelters and pig iron
- production (Scenario [3] in Fig. 2) clearly overestimated the contribution of emissions from these
- two sources at that time.

The ice-core ncCd trend indicates a steady increase from 1950 until the 1980s (Fig. 2b), 339 whereas past deposition calculated using Scenario [1] indicates a maximum in the mid-1960s 340 followed by a rapid decrease to 2000. Similarly, the ice-core ncZn peaks in the mid-1970s, 341 whereas calculated emissions peak in the mid-1960s. The 10-15 yr delay between the peak Cd 342 and Zn emissions in Scenario [1] and the ice-core data are likely a result of poorly constrained 343 EFs for Zn smelting, the dominant emissions source for both metals, compared to the much 344 better quantified decrease of Pb additives used in gasoline. Fig. 2 suggests that the EFs from Zn 345 smelters may have only decreased by a factor of 2 instead of 4-5 from ~1970 to 1980, a factor of 346 5 instead of 10 from ~1970 to 2000. 347

In conclusion, the observed changes of Pb, Cd, and Zn at CDD over recent decades indicates that the decrease of ice concentrations of Cd (consistent with Zn) is less than expected based on previous estimates of decreasing emissions factors after 1970, in particular from nonferrous metal smelting.

352 The calculated deposition fluxes at CDD indicate an increase by a factor of ~3 from 1890 to 1950 for anthropogenic Pb, Cd, and Zn (Fig. 2). This increase is comparable to the observed 353 increases of ncZn and ncCd ice concentrations (a factor of 3 and 2.2 from 1890 to 1950, 354 respectively), but is larger than the observed increase of ~1.5 for ncPb ice concentrations. At the 355 beginning of the 20th century, Pb emissions were dominated by Pb mining and smelting (42% of 356 total), and both Zn and Cd emissions were dominated by Zn smelters (Fig. S6). The difference 357 358 between the estimated Pb deposition and ncPb ice-core concentrations suggest that the EFs used at the beginning of the 20th century underestimate emissions by a factor of 2-3 with respect to the 359 1955 value. 360

At the beginning of the industrial period, the statistics indicate a more rapid increase of emissions for Zn and Cd (a factor of 3 from 1860 to 1885) than for Pb (a factor of 1.7), likely a result of significant Pb production in Spain in the mid-19th century (Table S4). The larger anthropogenic emissions of Pb compared to Cd and Zn at the very beginning of the industrial era are partly reflected in ice, with ncCd and ncZn concentrations near their pre-industrial (1700 to 1835) values in 1875, whereas ncPb were already 20 times above its pre-industrial baseline at that time (Table S3).

368 **5 Conclusions**

Continuous chemical measurements in alpine ice-cores were used to document preindustrial and year-by-year, 1890 to 2000 summer Cd and Pb pollution in western Europe, with ancillary measurements of Zn and Ce used for source apportionment. For the first time, alpine ice-core trends were evaluated in detail using new pollution estimates for the region based

- on country-specific emissions derived from updated source-category statistics, as well as state-
- of-the-art atmospheric aerosol transport and deposition modeling. Comparisons with existing
- pollution reconstructions back to 1955 show that while Pb pollution largely from leaded gasoline
- was similar to estimates, Cd pollution primarily from Zn smelter emissions was up to four-fold
- higher between the mid-1970s and 2000. In addition, peak Cd pollution occurred in the early1980s (i.e., almost coinciding with the Pb maximum) rather than the mid-1960s as previously
- thought. It is suggested that previous reconstructions of Cd pollution assumed overly optimistic
- emissions reductions from the introduction of pollution control devices and other technological
- improvements particularly between 1965 and 1980. This finding has implications for the impact
- of Cd pollution throughout ecosystems, which is sensitive to the concentration in the
- environment and also the duration of exposure.
- 384

385 Acknowledgments, Samples, and Data

The ice-core drilling operations at CDD were supported by the European Community via ENV4-

387 CT97 (ALPCLIM) contract and the Region Rhône-Alpes. The LEFE-CHAT (CNRS) program

entitled "Evolution séculaire de la charge et composition de l'aérosol organique au dessus de

³⁸⁹ l'Europe (ESCCARGO)" provided funding for analysis in France with the support of ADEME

390 (Agence de l'Environnement et de la Maîtrise de l'Energie). NSF grant 1925417 to JRM provided

³⁹¹ partial support for the analyses and interpretation at DRI. We thank colleagues who participated

in the drilling campaigns at CDD. We thank the two anonymous reviewers for their very usefulcomments, which improved the quality of the manuscript.

394 Ice-core data are available at NCEI (National Centers for Environmental Information) data base

- 395 (https://www.ncdc.noaa.gov/paleo/study/29492).
- 396

397 **References**

- Bowen, H. (1966), *Trace Elements in Biochemistry*, Academic Press, New York.
- Eckhardt, S., M. Cassiani, N. Evangeliou, E. Sollum, I. Pisso, and A. Stohl (2017), Source receptor matrix calculation for deposited mass with the Lagrangian particle dispersion
 model FLEXPART v10.2 in backward mode, *Geoscientific Model Development*, 10(12),
 4605-4618.
- Fagerli, H., M. Legrand, S. Preunkert, V. Vestreng, D. Simpson, and M. Cerqueira (2007),
 Modeling historical long-term trends of sulfate, ammonium, and elemental carbon over
 Europe: A comparison with ice core records in the Alps, *Journal of Geophysical Research*, 112, D23S13, doi:10.1029/2006JD008044.
- Gabrieli, J. (2008), Trace elements and polycyclic aromatic hydrocarbons (PAHs) in snow and
 ice sampled at Colle gnifetti, Monte Rosa (4450 m), during the past 10,000 years:
 environmental and climatic implications, 178 pp, University Ca' Foscari of Venice, Italy.

Gabrieli, J., and C. Barbante (2014), The Alps in the age of the Anthropocene: the impact of human activities on the cryosphere recorded in the Colle Gnifetti glacier, *Rendiconti Lincei-Scienze Fisiche E Naturali*, 25(1), 71-83.

413 414 415 416	Ilyin, I., O. Rozovskaya, O. Travnikov, M. Varygina, and W. Aas (2015), Heavy metals : Analysis of long-term trends, country-specific research and progress in mercury regional modelling. Convention on Long-range Transboundary Air Pollution, the European Monitoring and Evaluation Programme <i>Rep</i> ., Oslo, Norway.
417 418	Laloyaux, P., et al. (2018), CERA-20C: A Coupled Reanalysis of the Twentieth Century, Journal of Advances in Modeling Earth Systems, 10(5), 1172-1195.
419 420 421 422	Legrand, M., S. Preunkert, B. Jourdain, J. Guilhermet, X. Fain, I. Alekhina, and J. R. Petit (2013), Water-soluble organic carbon in snow and ice deposited at Alpine, Greenland, and Antarctic sites: a critical review of available data and their atmospheric relevance, <i>Climate of the Past</i> , <i>9</i> (5), 2195-2211.
423 424 425 426 427	 Legrand, M., J. R. McConnell, S. Preunkert, M. Arienzo, N. Chellman, K. Gleason, T. Sherwen, M. J. Evans, and L. J. Carpenter (2018), Alpine ice evidence of a three-fold increase in atmospheric iodine deposition since 1950 in Europe due to increasing oceanic emissions, <i>Proceedings of the National Academy of Sciences of the United States of America</i>, 115(48), 12136-12141.
428 429 430	McConnell, J. R., and R. Edwards (2008), Coal burning leaves toxic heavy metal legacy in the Arctic, <i>Proceedings of the National Academy of Sciences of the United States of America</i> , 105(34), 12140-12144.
431 432 433	McConnell, J. R., et al. (2018), Lead pollution recorded in Greenland ice indicates European emissions tracked plagues, wars, and imperial expansion during antiquity, <i>Proceedings of</i> <i>the National Academy of Sciences of the United States of America</i> , 115(22), 5726-5731.
434 435 436	McConnell, J. R., et al. (2019), Pervasive Arctic lead pollution suggests substantial growth in medieval silver production modulated by plague, climate, and conflict, <i>Proceedings of the National Academy of Sciences of the United States of America</i> , 116(30), 14910-14915.
437	Minemet statistical yearbooks, Minemet- Groupe Imetal, Paris, (1913–1986).
438	Mitchell, B. (1975), European historical statistics, 1750-1970, Palgrave Macmillan UK, UK.
439 440	Nriagu, J. O. (1979), A global assessment of natural sources of atmospheric trace metals, <i>Nature</i> , 338, 47-49.
441 442	Nriagu, J. O. (1990), Global metal pollution: Poisoning the biosphere?, <i>Environment: Science</i> and Policy for Sustainable Development, 32 (7), 7-33.
443 444	Nriagu, J. O., and J. M. Pacyna (1988), Quantitative assessment of worldwide contamination of air, water and soils by trace metals, <i>Nature</i> , <i>333</i> (6169), 134-139.
445 446 447	Olendrzyński, K., S. Anderberg, J. Bartnicki, J. Pacyna, and W. Stigliani (1996), Atmospheric emissions and depositions of cadmium, lead and zinc in Europe during the period 1955–1987, <i>Environmental Reviews</i> , 4(4), 300-320.
448 449 450	Pacyna, J. (1991), Emission factors of atmospheric Cd, Pb and Zn for major source categories in Europe during 1950-1985, Norwegian Institute for Air Research, Technical Report NILU-OR-30/91, 34 pp, Norway.

451	Pacyna, J. M., and E. G. Pacyna (1999), Atmospheric emissions of anthropogenic lead in
452	Europe: improvements, updates, historical data and projections, Technical report 2000/31
453	for GKSS Research Center, Geesthacht, Germany.
454 455 456	Pacyna, J. M., and E. G. Pacyna (2001), An assessment of global and regional emissions of trace metals to the atmosphere from anthropogenic sources worldwide, <i>Environmental Reviews</i> , 9(4), 269-298.
457 458	Pacyna, J. M., E. G. Pacyna, and W. Aas (2009), Changes of emissions and atmospheric deposition of mercury, lead, and cadmium, <i>Atmospheric Environment</i> , 43(1), 117-127.
459	Preunkert, S., Wagenbach, D., Legrand, M., and Vincent, C. (2000), Col du Dome (Mt Blanc
460	Massif, French Alps) suitability for ice-core studies in relation with past atmospheric
461	chemistry over Europe, <i>Tellus Series B-Chemical and Physical Meteorology</i> , 52(3), 993-
462	1012.
463 464 465	Preunkert, S., D. Wagenbach, and M. Legrand (2002), Improvement and characterization of an automatic aerosol sampler at high-Alpine drilling sites, <i>Atmospheric Environment</i> , 36, 1221-1232.
466	Preunkert, S., and M. Legrand (2013), Towards a quasi-complete reconstruction of past
467	atmospheric aerosol load and composition (organic and inorganic) over Europe since
468	1920 inferred from Alpine ice cores, <i>Climate of the Past</i> , 9(4), 1403-1416.
469	Schmitz, C. J. (1979), World non-ferrous metal production and prices, 1700-1976, Routledge.
470	Schwikowski, M. (2004), Reconstruction of European air pollution from Alpine ice cores, in
471	<i>Earth Palaeoenvironments: Records Preserved in Mid- and Low-Latitude Glaciers</i> ,
472	edited by L. D. Cecil, J. R. Green and L. G. Thompson, pp. 95-119, Kluwer Academic
473	Publishers, Dordrecht.
474	Schwikowski, M., et al. (2004), Post-17th-century changes of European lead emissions recorded
475	in high-altitude alpine snow and ice, <i>Environmental Science & Technology</i> , <i>38</i> (4), 957-
476	964.
477	Stohl, A., C. Forster, A. Frank, P. Seibert, and G. Wotawa (2005), Technical note: The
478	Lagrangian particle dispersion model FLEXPART version 6.2, <i>Atmospheric Chemistry</i>
479	and Physics, 5, 2461-2474.
480	U.S. Geological Survey statistics, National Minerals Information Center,
481	https://www.usgs.gov/centers/nmic/minerals-yearbook-metals-and-minerals
482	Vestreng, V., E. Rigler, M. Adams, K. Kindbom, J. M. Pacyna, H. D. van der Gon, S. Reis, and
483	O. Travnikov (2006), Emission Data Reported to the LRTAP Convention and NEC
484	Directive, Convention on Long-range Transboundary Air Pollution, the European
485	Monitoring and Evaluation Programme, Technical Report MSC-W 1/2006, Oslo,
486	Norway.
487 488 489 490	 Wagenbach, D., Preunkert, S., Schäfer, J., Jung, W. and Tomadin, L. (1996), Northward transport of Saharan dust recorded in a deep alpine ice core. In: Guerzoni, S. and Chester, R., (eds), <i>The Impact of Desert Dust Across the Mediterranean</i>. Kluwer Academic Publishers, Dordrecht, 291-300.

- Wagenbach, D., P. Bohleber, and S. Preunkert (2012), Cold, alpine ice bodies revisited: what
 may we learn from their impurity and isotope content?, *Geografiska Annaler Series a Physical Geography*, 94A(2), 245-263.
- Wagenbach, D., K. Münnich, U. Schotterer, and H. Oeschger (1988), The anthropogenic impact
 on snow chemistry at Colle Gnifetti, Swiss Alps, *Annals of Glaciology*, *10*, 183-187.
- 496
- 497

498

499 **Figure Captions:**

- 500 **Figure 1**. The 1890-2000 CDD ice records of non-crustal lead (a), cadmium (b), and zinc (c).
- 501 The grey lines refer to raw data (left scales). The red curves and blue symbols are the smoothed 502 summer profiles (first component of single spectra analysis with a 5-yr time window, 5yr-SSA)
- 502 summer promes (first component of single spectra analysis with a 3-yr till 503 and winter values, respectively (right scales)
- and winter values, respectively (right scales).
- **Figure 2**. a–c: Summer ice-core trends of non-crustal concentrations (black lines: 5yr- SSA) and anthropogenic deposition fluxes at CDD calculated using FLEXPART simulations and estimated
- past anthropogenic emissions from 1890 to 2000. Red squares and blue triangles refer to
- deposition fluxes obtained by using 1955-2000 emissions (see section 2.5), and 1890-1970
- statistical data with EFs for prior to 1970 (except for coal, Table S2), respectively. The dashed
- 509 blue line is calculated deposition after 1970 (except for coal, Table 52), respectively. The dashed
- 510 metallurgy as before 1970. d-f: Estimated summer deposition fluxes since 1890 related to the
- 511 different source categories and applying assumptions made for Scenario [3] (see section 3.2).
- 512 Blanks in the 1975-1979 and 1985-1991 years denote lack of statistical data.

513

Figure 1.



Figure 2.

